ORIGINAL RESEARCH



# Healthcare Professionals' Knowledge of and Attitudes Towards the Use of Time in Range in Diabetes Management: Online Survey Across Seven Countries

Christophe De Block ( · Alice Y. Y. Cheng · Trine Brandt Christensen · Usha Rani H. Patted · Anna Ginovker

Received: January 6, 2023 / Accepted: May 31, 2023 / Published online: June 18, 2023  $\odot$  The Author(s) 2023

## ABSTRACT

*Introduction*: Time in range (TIR) is a metric of glycaemic target management derived from continuous glucose monitoring (CGM) data. This study aimed to understand knowledge of and attitudes towards use of TIR among healthcare professionals (HCPs), and gain insights into benefits and barriers to its use in clinical practice.

*Methods*: An online survey was disseminated across seven countries. Participants were sampled from online HCP panels and were aware of TIR (defined as amount of time in, below, and

**Supplementary Information** The online version contains supplementary material available at https://doi.org/10.1007/s13300-023-01429-x.

C. De Block (🖂)

Laboratory of Experimental Medicine and Paediatrics, Member of the Infla-Med Center of Excellence, Faculty of Medicine and Health Science, University of Antwerp, Antwerp, Belgium e-mail: christophe.deblock@uza.be

#### C. De Block

Department of Endocrinology, Diabetology and Metabolism, Antwerp University Hospital, Faculty of Medicine and Health Sciences, University of Antwerp, Antwerp, Belgium

A. Y. Y. Cheng

Trillium Health Partners and Unity Health, University of Toronto, Toronto, ON, Canada above target range). Participants were HCPs classified as specialists (SP), generalists (GP), or allied HCPs (AP; diabetes nurse specialists, diabetes educators, general nurses, nurse practitioners/physician assistants).

**Results:** Respondents included 741 SP, 671 GP and 307 AP. Most HCPs (approximately 90%) agreed TIR is likely/somewhat likely to become the standard of diabetes management. Perceived benefits of TIR included helping to optimise medication regimen (SP, 71%; GP, 73%; AP, 74%), giving HCPs the knowledge and insights to make informed clinical decisions (SP, 66%; GP, 61%; AP, 72%), and empowering people with diabetes with information to successfully manage their diabetes (SP, 69%; GP,

T. B. Christensen Novo Nordisk A/S, Copenhagen, Denmark

*Present Address:* T. B. Christensen Ascendis Pharma, Copenhagen, Denmark

U. R. H. Patted Novo Nordisk, GBS, Bangalore, Karnataka, India

A. Ginovker The Harris Poll, Princeton, NJ, USA

C. De Block University Hospital Antwerp, Drie Eikenstraat 655, 2650 Edegem, Belgium 77%; AP, 78%). Barriers to wider adoption included limited CGM access (SP, 65%; GP, 74%; AP, 69%) and lack of HCP training/education (SP, 45%; GP, 59%; AP, 51%). Most participants considered integration of TIR into clinical guidelines, recognition of TIR by regulators as a primary clinical endpoint, and recognition of TIR by payers as a parameter for diabetes treatment evaluation as key factors for increased use.

*Conclusions*: Overall, HCPs agreed on the benefits of using TIR for diabetes management. Besides raising awareness among HCPs and people with diabetes, more training and healthcare system updates are needed to facilitate increased TIR use. In addition, integration into clinical guidelines and recognition by regulators and payers are needed.

## PLAIN LANGUAGE SUMMARY

'Time in range' is the proportion of time in a day that a person's glucose level is within a particular range. The purpose of this study was to understand knowledge of and attitudes towards use of TIR among healthcare professionals. The study was carried out using an online survey and participants from seven countries were included. Participants were healthcare professionals classified as specialists (SP), generalists (GP), or allied healthcare professionals (AP; diabetes nurse specialists, diabetes educators, general nurses, nurse practitioners, or physician assistants). Overall, 1719 participants were included in the study. Most healthcare professionals (approximately 90%) agreed that time in range is likely/somewhat likely to become the standard of diabetes management. Participants reported the following benefits of time in range: helping to optimise medication regimen, giving healthcare professionals the knowledge and insights to make informed clinical decisions, and empowering people with diabetes with information to successfully manage their diabetes. The most common barrier to wider time in range adoption was limited access to continuous glucose monitoring (SP, 65%; GP, 74%; AP, 69%), followed by lack of healthcare professionals' training/education (SP, 45%; GP, 59%; AP, 51%). Most participants considered integration of time in range into clinical guidelines, recognition of time in range by regulators as a primary clinical endpoint, and recognition of time in range by payers as a parameter for evaluation of diabetes treatment as key factors for the increased use of time in range.

**Keywords:** Clinical practice; Continuous glucose monitoring; Diabetes management; Glycaemic management; Time in range

#### **Key Summary Points**

#### Why carry out this study?

Time in range (TIR) is a metric of glycaemic target management derived from continuous glucose monitoring (CGM) data and can be complementary to the data provided by glycated haemoglobin (HbA1c).

TIR can be a useful tool in diabetes management by providing actionable insights that can improve glycaemic target management.

This study aimed to understand knowledge of and attitudes towards the use of TIR among healthcare professionals (HCPs), and to gain insights into the benefits and barriers to its use in clinical practice.

#### What was learned from the study?

Overall, HCPs agreed on the benefits of using TIR for diabetes management, with approximately 90% agreeing that it is likely/somewhat likely to become the standard of diabetes management.

However, there are barriers to the wider adoption of TIR in clinical practice, including limited access to CGM for people with diabetes, and lack of training and education for HCPs.

Our results demonstrate the need for further HCP education and training on TIR, incorporation of TIR into clinical guidelines, and recognition by regulators and payers in order to increase and optimise the use of TIR in clinical practice.

Although most HCPs who use ambulatory glucose profile (AGP) reports indicated that they are very comfortable or comfortable interpreting them, a sizeable minority indicated that they are only somewhat or not comfortable, suggesting a potential need for more training on the interpretation of AGP reports.

## INTRODUCTION

The assessment of glycaemic target management is central to informing and directing good diabetes management [1, 2]. For several decades, the gold standard for monitoring longterm glycaemic stability and assessing the risk of diabetes-related complications has been glycated haemoglobin (HbA1c), which provides an average of blood glucose levels over approximately the last 120 days [3–5]. However, HbA1c does not provide information on the magnitude and frequency of intra- and inter-day blood glucose fluctuations, nor does it reflect the risk of acute hypo- or hyperglycaemic events [6]. Furthermore, HbA1c may be an unreliable measure of glycaemic stability and variability in individuals with conditions that affect red blood cell lifespan, e.g. anaemia or certain haemoglobinopathies [7], and can vary across different ethnicities, e.g. HbA1c is 0.4% higher in African American and Hispanic people than in Caucasian people for the same mean glucose concentration [8].

Continuous glucose monitoring (CGM) devices measure glucose levels within the interstitial fluid continuously over a 24-h period [9]. Therefore, CGM enables healthcare professionals (HCPs) and people with diabetes to view more comprehensive daily glucose profiles and better assess glycaemic variability [10]. Guidelines recommend real-time CGM (rt-CGM) or intermittently scanned CGM (is-CGM) in adults with diabetes on multiple daily injections or continuous subcutaneous insulin infusion for management of their diabetes [11]. The use of CGM is now increasing in many countries, but differences between countries still exist depending on healthcare systems and resources [12, 13].

Time in range (TIR), time above range (TAR), and time below range (TBR) are emerging glycaemic metrics based on CGM data and can be complementary to the data provided by HbA1c [6, 11]. Guidance on target glucose ranges and recommendations for time spent within these ranges have been provided by an International Consensus on Time in Range (IC-TIR) for different diabetes populations [14]. For the majority of people with type 1 diabetes (T1D) or type 2 diabetes (T2D), > 70% of readings within the target range 70–180 mg/dl (3.9–10.0 mmol/ l) and < 4% below 70 mg/dL are recommended [14].

TIR has previously been evaluated as an outcome measure for diabetes clinical trials and has been shown to correlate with risk of complications. A study published in 2019 by Beck et al. demonstrated that the mean TIR (70-180 mg/dl [3.9-10.0 mmol/l]) derived from 7-point self-monitored blood glucose (SMBG) profiles had a strong association with the risk of development of retinopathy and microalbuminuria, and with progression of retinopathy. Using public data sets from the Diabetes Control and Complications Trial (DCCT), the authors determined that the adjusted hazard rates for development of retinopathy progression and of microalbuminuria significantly increased by 64% (95% CI 51-78) and 40% (95% CI 25-56), respectively, for each 10%point decrease in SMBG-derived TIR [15]. Several studies in which CGM data were used have demonstrated that lower TIR was associated with microvascular complications (presence of neuropathy, retinopathy, or nephropathy), and with hypoglycaemia or ketoacidosis [10, 14–17]. The use of TIR in clinical practice also offers the potential for a personalised approach to diabetes management [14, 18, 19].

To review implementation and optimise TIR use in clinical practice, it is important to understand current awareness and attitudes of HCPs to the metric. The aim of the current quantitative survey was to establish a benchmark for the current levels of knowledge and attitudes towards use of TIR among HCPs (specialists, generalists, and allied HCPs). In addition, we sought to survey the key drivers and barriers of using TIR in clinical practice with a view to identifying the support required for potential integration of TIR into clinical practice.

## **METHODS**

#### Study Design

This quantitative survey involved disseminating an online questionnaire across seven countries (Brazil, Canada, South Korea, Spain, Sweden, UK, and the USA) between 17 November 2021 and 4 February 2022. The country selection was based on providing international generalisability and representing countries that have shown early adoption of technology and therefore moderate-to-high use of CGM [20]. The survey was designed by The Harris Poll in partnership with Novo Nordisk and an expert review panel. The survey was hosted by The Harris Poll on a secure website and took approximately 20 min to complete. All HCPs who completed the survey received honorarium for participation that was in line with the fair market value in each of the countries.

HCPs were identified through online panels and targeted by specialty; the overall sampling was managed by The Harris Poll. All panellists had opted into receiving survey invitations and had agreed to take part in online research. Once identified, HCPs were invited to a secure website or member portal via email to be screened and, if they qualified, to complete an online self-administered questionnaire. Each invitation contained a unique password-protected survey link to ensure anonymity and prevent participants from completing the survey more than once.

#### Participants

To qualify for survey completion, participants had to have been in practice for at least 1 year, spend at least 70% of their time in direct patient care, and be aware of TIR as a metric for diabetes management. The 70% threshold for time in direct patient care was used to limit the proportion of HCPs who, although they may be knowledgeable about TIR, spend a considerable amount of time teaching and/or in academia. Participants in the study were HCPs classified as specialists, generalists, or allied HCPs. Classification was by participants self-identifying as part of the questionnaire and according to a

combination of different factors and responses to the screening questions (see Q1000 section of the questionnaire in the electronic supplementary material). Most specialists had their primary specialty in endocrinology or diabetology ('specialist' also included those whose primary specialty was family practice/general practice/primary care/internal medicine, who had a specialisation in endocrinology and saw at least 50 people with diabetes per month), and generalists had their primary specialty in family practice, general practice, primary care, or internal medicine and saw at least five people with diabetes. In Brazil and Sweden, cardiologists and geriatricians were classified as generalists, as they are often the first contact for people with diabetes with T2D. Allied HCPs were diabetes nurse specialists, diabetes educators, general nurses, nurse practitioners, or physician assistants and saw at least five people with diabetes. See Table S1 in the electronic supplementary material for the full criteria for each HCP subgroup.

### Survey Questionnaire

The survey captured characteristics of the participants, including years in practice (there was no upper limit for years in practice), location and setting of their practice, average duration of time spent in direct patient care, and number of people with diabetes seen per month. Statements regarding the importance, potential benefits, and impact of TIR, as well as barriers to adoption of TIR, were also included in the questionnaire, which is available in the online supplementary material. For the purpose of this study, people with diabetes were defined as adults aged  $\geq$  18 years with T1D or T2D, CGM was defined as rt-CGM (e.g. Dexcom G5 or G6, Guardian 3, Enlite, Eversense) or is-CGM (e.g. FreeStyle Libre 1, 2, or Pro), and TIR was defined as the amount of time spent in range, below range, and above range.

#### **Statistical Methods**

Results were not weighted and are only representative of HCPs who participated in the study. Data were collected and analysed by HCP subgroup: specialists, generalists, or allied HCP, and statistical testing was performed to compare differences in responses between these subgroups. These differences were tested at the 95% level of confidence. The *t* test of means and *Z* test of proportions were used. No multiple comparison adjustments were made.

#### **Compliance with Ethics Guidelines**

Participants provided consent to take part in the study and were given the option to waive confidentiality in the event of any safety or adverse event information reported to the sponsor's pharmacovigilance department. The study was performed in accordance with the Helsinki Declaration of 1964. This study adhered to the EphMRA Code of Conduct 2019, page 15, section 1.3.

## RESULTS

## Participant Characteristics

In total, 1719 participants were included in the study: 741 specialists, 671 generalists, and 307 allied HCPs. The profiles of the three HCP subgroups are presented in Table 1. In each HCP subgroup, the majority of participants were based at urban practices, with the proportion being slightly higher in the specialist subgroup (Table 1). A numerically greater proportion of generalists were based in rural practices compared with specialist or allied HCPs (Table 1). The mean number of years in practice was similar across the three HCP subgroups (17–20 years) (Table 1).

## **Current Perception of TIR**

Approximately 90% of participants believed that TIR is very or somewhat likely to become the standard of diabetes management; this was

	Specialists (n = 741)	Generalists (n = 671)	Allied HCPs $(n = 307)$
Country, <i>n</i> (%)			
Brazil	142 (19)	100 (15)	51 (17)
Canada	92 (12)	131 (20)	50 (16)
South Korea	100 (13)	120 (18)	50 (16)
Spain	120 (16)	101 (15)	32 (10)
Sweden	35 (5)	29 (4)	15 (5)
UK	151 (20)	50 (7)	47 (15)
USA	101 (14)	140 (21)	62 (20)
Practice location, $n$ (%)	(n = 599)	(n = 571)	(n = 256)
Urban	431 (72)	357 (63)	167 (65)
Suburban	136 (23)	146 (26)	66 (26)
Rural	32 (5)	68 (12)	23 (9)
Practice setting, n (%)	(n = 599)	(n = 571)	(n = 256)
Family/general/IM practice (single specialty)	106 (18)	448 (78)	44 (17)
Endocrinology/diabetology practice (single specialty)	179 (30)	20 (4)	45 (18)
Multi-specialty practice	152 (25)	71 (12)	60 (23)
Hospital	151 (25)	21 (4)	82 (32)
Community practice (UK only)	9 (2)	2 (0)	13 (5)
Other	2 (0)	9 (2)	12 (5)
Duration in practice, mean (SD), years	17 (7)	20 (9)	18 (10)
Proportion of time spent in direct patient care, mean (SD), %	90 (9)	92 (9)	87 (11)
Mean number of people with diabetes seen per month, $n$			
T1D	54	16	38
T2D	198	141	106
Mean proportion of people with diabetes using CGM, %			
TID	(n = 735)	(n = 630)	(n = 292)
	49	45	52
T2D	(n = 739)	(n = 671)	(n = 307)
	23	24	31
T2D not taking insulin	(n = 739)	(n = 671)	(n = 307)
	10	11	13

#### Table 1 Characteristics of participants

CGM continuous glucose monitoring, HCP healthcare professional, IM internal medicine, n number of participants, SD standard deviation, T1D type 1 diabetes, T2D type 2 diabetes



(c) Participants rating TIR as very effective/effective for optimising non-insulin regimens for people with T2D



(b) Participants rating TIR as very

effective/effective for optimising insulin regimen

83

87

100

80

60

40

87\*

(d) Participants who identify the ability of TIR to empower HCPs with the knowledge and insights they need to make informed clinical decisions as its main benefit



Fig. 1 Value of TIR for clinical decision-making, as reported by HCPs. \*p < 0.05 for difference versus generalists. HCP healthcare professional, T2D type 2 diabetes, TIR time in range

reported for specialists (93%; p < 0.05 vs. generalists), allied HCPs (96%; p < 0.05 vs. specialists and generalists), and generalists (87%).

Participants cited a range of benefits to using TIR for clinical decision-making. The ability to help optimise medication regimens was identified as a main benefit of using TIR by 71–74% of participants (Fig. 1). More than 8/10 participants reported finding the use of TIR very

effective or effective for optimising insulin regimens, and at least one in two considered it very effective or effective for optimising non-insulin regimens for people with T2D, which was reported particularly among specialists and allied HCPs, compared with generalists (p < 0.05 for both comparisons) (Fig. 1).

At least 60% of participants reported the empowering of HCPs with the knowledge and



Fig. 2 Value of TIR for empowering people with diabetes to manage their diabetes, as reported by HCPs. \*p < 0.05 for difference versus specialists. *HCP* healthcare professional, *TIR* time in range

insights they need to make informed clinical decisions to be a main benefit of TIR, and approximately 90% strongly or somewhat agreed that it makes them feel more confident in managing people with diabetes (Fig. 1).

In addition to empowering HCPs, the majority of participants emphasised that TIR empowers the person with the information they need to successfully manage their diabetes, enhances their self-management, and lowers their risk of complications by helping to improve their blood glucose levels (Fig. 2). Empowering people with diabetes with information and lowering the risk of complications were reported as benefits by a greater proportion of allied HCPs than specialists (p < 0.05 vs. specialists) (Fig. 2).

TIR can be used as a tool to inform lifestyle modifications, with 76% of specialists, 73% of generalists, and 82% of allied HCPs (p < 0.05 vs.

specialists and generalists) considering it to be very effective or effective (Fig. S1a in the electronic supplementary material). Furthermore, the majority of participants agreed that TIR allows people with diabetes to immediately see the direct impact of their actions (Fig. S1b). The majority of participants considered TIR to be very effective or effective in providing accurate information about hypoglycaemia, and strongly or somewhat agreed that TIR helps to reduce patient fear of hypoglycaemia and provides a more accurate and complete picture of glucose fluctuations than HbA1c (Fig. S1c, d).

From the perspective of at least half of the participants, TIR provides a more accurate and complete picture of glucose fluctuations than HbA1c, offers the benefit of facilitating more effective patient–physician dialogue, helps save time for both people with diabetes and physicians, and makes it easier to manage people





Fig. 3 a Knowledge of and **b** comfort with using TIR, and **c** use of AGP reports, among participants. \*p < 0.05 for difference versus generalists and allied HCPs. †p < 0.05

with diabetes virtually (Fig. S1e–h). These benefits were reported by a greater proportion of allied HCPs than specialists and generalists.

# Current Knowledge and Use of TIR by HCPs

Knowledge and comfort with using TIR as well as use of ambulatory glucose profile (AGP) reports among participants were limited, particularly among generalists compared with specialists and allied HCPs (Fig. 3). The proportion of participants reporting that they were very comfortable/comfortable using TIR was 76% for specialists (p < 0.05 vs. generalists and allied HCPs), 42% for generalists, and 63% for allied HCPs (p < 0.05 vs. generalists).



(b) Patients who are very comfortable or comfortable with using TIR for diabetes management

for difference versus generalists. *AGP* ambulatory glucose profile, *HCP* healthcare professional, *TIR* time in range

# Training Requirements and Barriers to Adoption of TIR

Overall, 32% of specialists (p < 0.05 vs. generalists), 11% of generalists, and 35% of allied HCPs (p < 0.05 vs. generalists) reported receiving formal training or instructions on TIR. Lack of TIR training/education for HCPs was considered a main barrier for wider adoption of TIR for diabetes management. The different aspects of training deemed to be required for TIR use by each of the three HCP subgroups are presented in Fig. 4. A greater proportion of generalists and allied HCPs versus specialists reported the need for the majority of training aspects. Training needs reported included clinical guidance about TIR targets for different patient populations, how to use TIR data to make clinical decisions, and how to interpret TIR data (Fig. 4).

Beyond their own training needs, almost all participants agreed that patient education on TIR is also necessary to increase adoption of the



Fig. 4 Training needs for TIR use, as reported by specialists, generalists, and allied HCPs. \*p < 0.05 for difference versus specialists. †p < 0.05 for difference versus

metric, with 93% of specialists (p < 0.05 vs. generalists), 89% of generalists, and 95% of allied HCPs (p < 0.05 vs. generalists) responding that patient education is absolutely essential, very important, or important. Overall, the majority of all participants strongly or somewhat agreed that there is a need for more resources to help educate people with diabetes on TIR. Limited access to CGM was deemed a barrier to wider adoption of TIR by 65% of specialists, 74% of generalists (p < 0.05 vs. specialists), and 69% of allied HCPs.

# Factors Necessary for Facilitating the Increased Use of TIR

Most participants considered integration of TIR into clinical guidelines to be a key factor for the increased use of TIR (Fig. 5). Raising awareness of TIR among HCPs and people with diabetes was also cited as important for the increased use of TIR (Fig. 5).

Recognition of TIR by regulators as a primary clinical endpoint was cited as a key factor for the increased use of TIR by 52–59% of participants (Supplementary information: the questionnaire, question 420). Finally, recognition of TIR by payers as a parameter for evaluation of diabetes treatment was cited as a key factor by the majority of participants (60–70%),

specialists and allied HCPs.  ${}^{\ddagger}p < 0.05$  for difference versus specialists and generalists. *HbA1c* glycated haemoglobin, *HCP* healthcare professional, *TIR* time in range

especially among allied HCPs (p < 0.05 for difference vs. specialists) (Fig. 5).

## DISCUSSION

TIR is a recent metric that complements the traditional measure of HbA1c [19]. In this study, the knowledge of and attitudes towards the use of TIR in diabetes management among HCPs with prior awareness of the metric were investigated via an online quantitative survey. Overall, most of the HCPs surveyed agreed that TIR is likely to become the standard metric for diabetes management in the future. TIR was regarded as providing many benefits for HCPs, including helping to optimise a patient's medication regimen and giving HCPs the knowledge and insights they need to make informed clinical decisions. In addition, the majority of participants felt that TIR empowers people with diabetes with the information they need to successfully manage their diabetes. However, the level of comfort with using TIR reported by HCPs was low, particularly for generalists, with lack of education or training for both HCPs and people with diabetes identified as a key barrier to wider adoption of TIR in practice. Additional barriers to implementation of TIR were identified which, if targeted, may increase awareness



Fig. 5 Factors necessary for facilitating the increased use of TIR. \*p < 0.05 for difference versus specialists.  $^{\dagger}p < 0.05$  for difference versus specialists and generalists. *HCP* healthcare professional, *TIR* time in range

and use of TIR, and ultimately help improve diabetes management for people with diabetes.

The results of the survey were analysed by HCP subgroup: specialists, generalists, and allied HCPs. The proportion of generalists expressing that TIR is becoming the standard of care in diabetes was slightly smaller than that of specialists and allied HCPs. This is likely to be due to generalists seeing fewer people with diabetes receiving insulin (and therefore possibly using CGM) than the other HCPs. Several benefits of TIR, including use as a tool to inform lifestyle modifications, facilitating more effective patient–physician dialogue, and saving time for both people with diabetes and HCPs, were reported by a significantly greater proportion of allied HCPs than by specialists or generalists. A greater proportion of allied HCPs than specialists reported empowering people with diabetes with information and lowering the risk of complications as benefits of TIR. This discrepancy may be due to the greater frequency of patient interactions that allied HCPs have compared with specialists. However, another possible explanation is that specialists may be more critical of potential benefits compared with allied HCPs. Despite these benefits, use of TIR and comfort level with using TIR were generally low for all HCPs, and particularly for generalists.

Furthermore, although HCPs' opinion on the need for further training and education on AGP reports was not directly questioned in the survey, the authors feel that education on how to interpret AGP reports (Supplementary information: the questionnaire, question 345), taking into account recommendations of > 70% TIR and < 4% TBR, is required. Indeed, a sizeable minority of HCPs were only somewhat comfortable or not comfortable with AGP interpretation and the explanation to people with diabetes; the results from the survey suggest that even HCPs who use AGP reports are not comfortable interpreting them. Interpreting trends (e.g. postprandial highs or nocturnal highs/lows) and how to address them also need to be discussed. All HCP subgroups identified the need for clinical guidance on TIR targets for different patient populations as important. However, the topics directed towards interpretation and use of TIR in clinical practice were emphasised more by generalists, which corresponds with the lower awareness, knowledge, and use of TIR and AGP in this subgroup. In the opinion of the authors, training should include structured interpretation of TIR data and guidance on the actions that should be taken in response to these data. For example, if TIR is < 70%, the HCP should check TBR and indicators of glycaemic variability. If TBR is > 4%, the timings of hypoglycaemic episodes should be assessed. Any TAR should be investigated to see if this occurs either postprandially or overnight. Training should also focus on enabling HCPs to articulate the benefits of TIR to their patient and on how to identify patterns in the data and discuss these with the patient.

In addition to HCP training on TIR, wider adoption of the metric is likely to be facilitated by its integration into clinical guidelines and recognition by regulators and payers. A consensus report published in 2019 by Battelino et al. provides recommendations on targets for TIR, TAR, and TBR [14]. Although this consensus report was endorsed by the European Association for the Study of Diabetes (EASD) and the American Diabetes Association (ADA) (among other bodies), integration of TIR into official clinical practice guidelines is likely to increase confidence of HCPs in its use and enable wider adoption. It will also be important to gain acceptance from regulators of TIR as a primary clinical endpoint for use in future trials. This will require expert HCPs participating in the preparation of consensus papers summarising the current evidence to date from trials using TIR as a clinical endpoint.

As mentioned, use of CGM can be dependent on reimbursement [12]. For more widespread reimbursement of CGM for people with diabetes wishing to use TIR, recognition of TIR as a parameter for evaluation of diabetes management by payers was considered a barrier to using TIR by the majority of participants in this study (60–70%).

The strengths of this study include an international design and assessing HCPs across seven countries on four continents. Alongside geographical range, three different HCP groups were analysed in each country, and so data generated take into account both degree of specialisation and HCP role. Consequently, the level of granular information detailed here should aid the development of tailored TIR training for HCPs.

As with all survey studies, our study has limitations. Reporting bias may have manifested, for example, as a result of HCPs being hesitant to admit not being comfortable with using a particular technique. This may have been the reason for the consistently lower proportion of specialists reporting need for TIR training compared with generalists and allied HCPs. Inclusion in this study required the HCP to be aware of TIR (although not necessarily to have experience in the practical use of TIR) and this requirement would have introduced bias. This study focused on global data and any future secondary publications could discuss the differences among countries.

## CONCLUSIONS

TIR as a metric is changing the way in which diabetes is managed. The majority of HCPs who took part in this survey recognised the value of TIR in aiding clinical decision-making and empowering people with diabetes in managing their diabetes. However, barriers to wider implementation of TIR exist. Our results identified that education is needed to increase awareness of TIR and to optimise its use in clinical practice. There is a need for considering TIR in clinical guidelines and for recognition by regulators and payers. Physicians and clinical managers should aim to review healthcare services with tailored support and training to help evaluate the use of TIR in clinical practice.

## ACKNOWLEDGEMENTS

Authors wish to thank the participants of the study. People with diabetes reviewed the HCP questionnaire in September 2021. The expert review panel included patient advocates who provided input into the survey design. No people with diabetes were asked to advise on interpretation or writing up of results. There are no plans to disseminate the results of the research to study participants or the relevant patient community.

*Funding.* This study and the journal's Rapid Service Fee were funded by Novo Nordisk A/S.

*Authorship.* All named authors meet the International Committee of Medical Journal Editors (ICMJE) criteria for authorship for this article, take responsibility for the integrity of the work as a whole, and have given their approval for this version to be published.

*Medical Writing, Editorial, and Other Assistance.* The authors would like to thank Roy Beck for the expertise that greatly assisted the research and for reviewing the questionnaire prior to its administration, and Yogesh Madhukar Kothawade for the planning and execution of the research study and facilitating local approval. Medical writing and editorial support for the development of this manuscript, under the direction of the authors, were provided by Matthew Robinson and Helen Marshall, of Ashfield MedComms, an Inizio company, and funded by Novo Nordisk.

*Author Contributions.* Christophe De Block, Alice YY Cheng, Trine Brandt Christensen, Usha Rani H Patted and Anna Ginovker contributed to the study conception and design. Material preparation (questionnaire finalisation) and data analysis were performed all authors. Data collection was led by Anna Ginovker. All authors contributed to critically reviewing and editing the manuscript drafts, and all authors read and approved the final manuscript.

**Prior Presentation.** This article extends observations previously presented at the EASD (2022) European Association for the Study of Diabetes—58th Annual Meeting (19–23 Sept 2022, Stockholm, Sweden).

Disclosures. Christophe De Block declares he has participated in advisory boards for Abbott Diagnostics, A. Menarini Diagnostics, Insulet, Roche Diagnostics, Eli Lilly and Company, Novo Nordisk, and Novartis; has received search support from AstraZeneca, Boehringer Ingelheim, Indigo NV, and Novo Nordisk; and declares participation in speaker bureaus for Eli Lilly and Company, Novo Nordisk, A. Menarini Diagnostics, Abbott Diagnostics, and Insulet. Alice YY Cheng has received honoraria for consulting and/or speaking from Abbott, Amgen, AstraZeneca, Bausch, Bayer, Boehringer Ingelheim, Dexcom, Eisai, Eli Lilly, GSK, Insulet, HLS Therapeutics, Janssen, Novo Nordisk, Pfizer, Sanofi, and Takeda; and has been involved in clinical trials with Applied Therapeutics and Sanofi. Trine Brandt Christensen and Usha Rani H Patted are employees of Novo Nordisk. Anna Ginovker is an employee of The Harris Poll, Princeton, NJ, United States.

*Compliance with Ethics Guidelines.* Participants provided consent to take part in the study and were given the option to waive confidentiality in the event of any safety or adverse event information reported to the sponsor's pharmacovigilance department. The study was performed in accordance with the Helsinki Declaration of 1964. This study adhered to the EphMRA Code of Conduct 2019, page 15, Sect. 1.3.

*Data Availability.* The datasets generated during and/or analysed during the current study are available from the corresponding author on reasonable request.

Open Access. This article is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License, which permits any non-commercial use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/bync/4.0/.

## REFERENCES

- American Diabetes Association Professional Practice Committee. 6. Glycemic targets: standards of medical care in diabetes-2022. Diabetes Care. 2022;45: S83–S96.
- 2. Davies MJ, D'Alessio DA, Fradkin J, et al. Management of hyperglycaemia in type 2 diabetes, 2018. A consensus report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetologia. 2018;61: 2461–98.

Tahara Y, Shima K. The response of GHb to stepwise plasma glucose change over time in diabetic

3.

4. Nathan DM, DCCT/EDIC Research Group. The diabetes control and complications trial/epidemiology of diabetes interventions and complications study at 30 years: overview. Diabetes Care. 2014;37: 9–16.

patients. Diabetes Care. 1993;16:1313-4.

- 5. Diabetes Control and Complications Trial Research Group, Nathan DM, Genuth S, et al. The effect of intensive treatment of diabetes on the development and progression of long-term complications in insulin-dependent diabetes mellitus. N Engl J Med. 1993;329:977–86.
- 6. Danne T, Nimri R, Battelino T, et al. International consensus on use of continuous glucose monitoring. Diabetes Care. 2017;40:1631–40.
- Beck RW, Connor CG, Mullen DM, Wesley DM, Bergenstal RM. The fallacy of average: how using HbA1c alone to assess glycemic control can be misleading. Diabetes Care. 2017;40:994–9.
- 8. Bergenstal RM, Gal RL, Connor CG, et al. Racial differences in the relationship of glucose concentrations and hemoglobin A1c levels. Ann Intern Med. 2017;167:95–102.
- 9. Petrie JR, Peters AL, Bergenstal RM, Holl RW, Fleming GA, Heinemann L. Improving the clinical value and utility of CGM systems: issues and recommendations: a joint statement of the European Association for the Study of Diabetes and the American Diabetes Association Diabetes Technology Working Group. Diabetes Care. 2017;40: 1614–21.
- Chehregosha H, Khamseh ME, Malek M, Hosseinpanah F, Ismail-Beigi F. A view beyond HbA1c: role of continuous glucose monitoring. Diabetes Ther. 2019;10:853–63.
- 11. American Diabetes Association Professional Practice Committee. 7. Diabetes technology: standards of medical care in diabetes-2022. Diabetes Care. 2022;45:S97–S112.
- DeSalvo DJ, Noor N, Xie C, et al. Patient demographics and clinical outcomes among type 1 diabetes patients using continuous glucose monitors: data from T1D Exchange real-world observational study. J Diabetes Sci Technol. 2021: 19322968211049783.
- 13. Dos Santos TJ, Dave C, MacLeish S, Wood JR. Diabetes technologies for children and adolescents with type 1 diabetes are highly dependent on coverage and reimbursement: results from a worldwide

survey. BMJ Open Diabetes Res Care. 2021;9: e002537.

- 14. Battelino T, Danne T, Bergenstal RM, et al. Clinical targets for continuous glucose monitoring data interpretation: recommendations from the International Consensus on Time in Range. Diabetes Care. 2019;42:1593–603.
- 15. Beck RW, Bergenstal RM, Riddlesworth TD, et al. Validation of time in range as an outcome measure for diabetes clinical trials. Diabetes Care. 2019;42: 400–5.
- 16. El Malahi A, Van Elsen M, Charleer S, et al. Relationship between time in range, glycemic variability, HbA1c, and complications in adults with type 1 diabetes mellitus. J Clin Endocrinol Metab. 2022;107:e570–81.
- 17. Lu J, Ma X, Zhou J, et al. Association of time in range, as assessed by continuous glucose

monitoring, with diabetic retinopathy in type 2 diabetes. Diabetes Care. 2018;41:2370–6.

- 18. TIRhub.com. How to use time in range in clinical practice. 2022. https://tirhub.com/how-to-use-tir-in-clinical-practice/. Accessed 1 June 2022.
- 19. Wilmot EG, Lumb A, Hammond P, et al. Time in range: a best practice guide for UK diabetes healthcare professionals in the context of the COVID-19 global pandemic. Diabet Med. 2021;38: e14433.
- Md Ismail Hossain, Ahmad Fadhil Yusof, Ab Razak Che Hussin, et al. Factors influencing adoption model of continuous glucose monitoring devices for internet of things healthcare. Internet of Things. 2021. https://wlv.openrepository.com/ bitstream/handle/2436/623851/Hossain\_et\_al\_ Factors\_influencing\_adoption\_model. pdf?sequence=3&isAllowed=n. Accessed 2 May 2023.